Author's response to reviews

Title: Inequality in oral health related to early and later life social conditions: a study of elderly in Norway and Sweden

Authors:

Ferda Gülcan (Ferda.Gulcan@iko.uib.no)
Gunnar Ekbäck (gunnar.ekback@orebroll.se)
Sven Ordell (Sven_ordell@hotmail.com)
Stein Atle Lie (Stein.Lie@iko.uib.no)
Anne Nordrehaug Åstrøm (Anne.Aastrom@iko.uib.no)

Version: 5 Date: 30 January 2015

Author's response to reviews:

Editor in chief
BMC Oral Health

Thank you for the review of the manuscript “Inequality in oral health related to early and later life social conditions: a study of elderly in Norway and Sweden”. We are grateful for the valuable comments from the reviewers and the editor. We have revised the manuscript according to the latest suggestions. Please find in the attachments the revised manuscript.

On behalf of the co-authors

Sincerely,

Ferda Gülcan